\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\AtBeginDocument{\let\citet\cite\let\citep\cite}
\usepackage[numbers,sort&compress,numbers,sort&compress]{natbib}



\newif\ifnokeywords\nokeywordsfalse
\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jrnlurlname{fjphs}
\usepackage{balance}

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\let\@eISSN\@empty
\let\@pISSN\@empty
\def\eISSN#1{\gdef\@eISSN{#1}}
\def\pISSN#1{\gdef\@pISSN{#1}}
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}

\usepackage{qrcode}

%\def\abbrvJournalTitle{Future J. Pharm. Health. Sci.}
\def\jshort{fjphs}

\def\title#1{%
  \gdef\@title{%
%      \vspace*{-20pt}%
       \vspace*{-48pt}
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%
         \parbox{.1\linewidth}{\includegraphics[width=4.2cm]{pharmasprings_logo.png}}\hfill
%         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.729\linewidth}{%
          \begin{center}
          \vskip-4pt 
          \scshape\fontsize{18}{20}\selectfont\journalTitle\hfill\end{center}%
          \vspace*{-16pt}{\vspace*{-3.1pt}\fontsize{8}{9}\selectfont Published by Pharma Springs Publication\hfill  Journal Home Page: \href{https://pharmasprings.com/index.php/\jrnlurlname}{\color{blue}\underline{\smash{https://pharmasprings.com/\jrnlurlname}}}}\hspace*{4pt}\mbox{}}}%
       \par\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}


\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}


                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
\def\bibjtitle#1{#1}
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}
\lefthyphenmin = 3
\makeatother
\date{}
\makeatletter
\usepackage{mdframed}
%\usepackage{balance}
\definecolor{unitednationsblue}{RGB}{222,234,246}
\definecolor{cpboxcolor}{RGB}{211,211,211}
\definecolor{citeboxcolor}{RGB}{242,242,242}
\def\printBio{%
  %\noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
  \noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerrightmargin=20pt,rightmargin=4,innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
   \@tempcnta=0
    \loop
      \advance \@tempcnta by 1
      \def\aubioCnt{\the\@tempcnta}
      \vskip1pt
%      \noindent\hspace{-2pt}%\fbox{
%     \vspace*{1pt}
      \raisebox{62pt}%{\begin{minipage}[t]{.35\linewidth}%
      \expandafter\ifx\csname authorImage\aubioCnt\endcsname\relax%
       \else%
         \includegraphics[height=60pt,width=60pt]{\csname authorImage\aubioCnt\endcsname}%height=145pt
       \fi%
      %\end{minipage}
      %}%}
      \hskip 4pt%\fbox{
      \raisebox{52pt}{\begin{minipage}[t]{.74\linewidth}%
        {\fontsize{8}{10}\selectfont{\csname authorName\aubioCnt\endcsname\\[5pt]}}
        \fontsize{8}{10}\selectfont\csname authorDetails\aubioCnt\endcsname\\
      \end{minipage}}
%      }
%     \vskip10pt%
    \ifnum\@tempcnta < \theaubio
    \repeat
%     \vspace*{-16.5pt}%
  \end{mdframed}
}


\makeatother


\def\citeauthor{~Muniraja~Lakshmi~K,
              ~Kiran~M,
              ~Sai~Prasanna~K,
              ~Appa~Rao}
\def\citetitle{Formulation and Characterization of Silver Nanoparticles Loaded with Aqueous Extract of \textit{Lantana Camara } Linn Leaves\textbf{\space }}
\def\copyrightstmt{This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial- ShareAlike 4.0 License, which allows others to remix, tweak, and build upon the work non-commercially, as long as the author is credited and the new creations are licensed under the identical terms.}
\publishedDate{}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Feature enabled:
%citation-style: unsrt
%citation-type: numbered
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[flushleft]{threeparttablex}

\usepackage{float}

\begin{document}


\def\RunningAuthor{Muniraja Lakshmi K et al.}
\firstPage{63}
\articleType{Original Article}
\receivedDate{20 Feb 2021}
\acceptedDate{06 Mar 2021}
\revisedDate{03 Mar 2021}
\journalVolume{1}
\journalIssue{2}
\journalDoi{fjphs.v1i2.248}
\copyrightYear{2021}
\eISSN{2583-116X}
\journalDoi{fjphs.v1i2.248}
\def\authorCount{4}
\def\affCount{3}

\def\journalTitle{Future Journal of Pharmaceuticals and Health Sciences}

\title{Formulation and Characterization of Silver Nanoparticles Loaded with Aqueous Extract of \textit{Lantana Camara } Linn Leaves\textbf{\space }}
\author{Muniraja~Lakshmi~K\textsuperscript{*}\textsuperscript{1},  
            Kiran~M\textsuperscript{2},  
            Sai~Prasanna~K\textsuperscript{2}, 
            Appa~Rao\textsuperscript{3}~\\[5pt]{\textsuperscript{1}Department of Pharmaceutics, Sri Venkateswara University of pharmaceutical sciences, Tirupati -517501, Andhra Pradesh, India}~\\{\textsuperscript{2}Department of Pharmacology, Sri Venkateswara University of pharmaceutical sciences, Tirupati-517501, Andhra Pradesh, India}~\\{\textsuperscript{3}Department of Biochemistry, Sri Venkateswara University of pharmaceutical sciences, Tirupati-517501, Andhra Pradesh, India}}

\begin{abstract}
In the present study, the \textit{In vitro} antibacterial activity and \textit{In vivo} anti-bacterial, anti-inflammatory, and wound healing activity of the leaf extract in solvents ethanol and aqueous extracts of the selected plant \textit{Lantana camara}. The synthesis of silver nanoparticles using aqueous extract of fresh leaves of \textit{Lantana camara} as bio reducing agents. This method allowed the synthesis of nanoparticles which was confirmed by UV-Visible spectroscopy, FTIR, particle size, and zeta potential. Anti-bacterial activity of ethanol and aqueous leaf extract of \textit{Lantana camara} was separately tested for gram-positive bacteria \textit{Staphylococcus aureus } and gram-negative bacteria \textit{Escherichia coli}. Again anti-bacterial activity of \textit{Lantana camara} mediated synthesis silver nanoparticles was tested by disc diffusion assay against standard organisms like \textit{Escherichia coli} and \textit{staphylococcus aureus} bacteria and \textit{in-vivo} studies designed to evaluate the carrageenan-induced paw edema and Excision wound model activities of ethanol and aqueous extract of leaf of \textit{Lantana camara} Linn. The ethanol and aqueous extracts of leaves give positive for all the phytochemical constituents etc. The particle size of silver particles prepared in all conditions was in the range of 2657.2nm. The Zeta potential of silver nanoparticles was found to be 0.3mv so it indicates the dispersion and stability.  
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Lantana Camara Leaves,\newline Phytochemical Constituents,\newline Silver Nanoparticles,\newline Anti Bacterial Activity,\newline Anti Inflammatory,\newline Wound Healing Activity
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright{\Large$^\ast$}Corresponding Author\par Name:\ Muniraja~Lakshmi~K~\\ Phone:\ +91 77998 71383~\\  Email:\ munirajalakshmi005@gmail.com
        \par}\vspace*{-13pt}\hrulefill\par\ifx\@eISSN\@empty\else{\fontsize{12}{14}\selectfont eISSN: \@eISSN\hfill pISSN: \@pISSN}\par\fi% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-12pt}\hrulefill\\[5pt]\hypersetup{allcolors=black}\qrcode[version=4,height=1.5cm]{https://doi.org/10.26452/\@journalDoi}\hypersetup{allcolors=blue} \hspace*{10.5pt}\parbox[t]{5cm}{\raisebox{19pt}{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\\[-13pt]{{\fontsize{12}{14}\selectfont  Pharmasprings.com \\[-1pt]{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\\[-8pt]\hspace*{-1pt}\rule{.762\columnwidth}{1.2pt}}}}\hypersetup{allcolors=blue}%{\fontsize{12}{14}\selectfont Pharmasprings.com}\par%
        \makeatother
        
    
\section{Introduction}
Nanoparticles represent a particle with a nanometer size of 1{\textendash}100 nm. The most widely studied nanoparticle materials are metal nanoparticles because they are easier to synthesize \unskip~\cite{1021853:21553581}. In the present work, \textit{Lantana camara} (also known as red sage) leaf extract was used to synthesize silver nanoparticles for antimicrobial studies. Some of the most studied metallic nanoparticles include silver, gold, platinum, and palladium. Ag nanoparticle is an interesting metal to be studied, especially in the field of health and medicine. Silver is a strong antibacterial and also toxic to cells \unskip~\cite{1021853:21553589}. Silver has the ability to damage bacterial cell walls, inhibits bacterial cell growth, and disrupts cell metabolism because of the interaction between Ag ions with macromolecules in cells, such as proteins and deoxyribonucleic acid. The Ag nanoparticles are chemically more reactive than silver in their bulk. The plant \textit{Lantana camara} Linn. (Verbenaceae) is an ornamental herb. It is found in South India, America, and Africa, mostly native to subtropical and tropical countries. The number of species is available with genus \textit{Lantana}. Some species of \textit{Lantana} are \textit{Lantana trifolia}, \textit{Lantana salvifolia} Jacq. and \textit{Lantana indica} Roxb. These leaves are rich in various chemical constituents such as triterpenes, glycosides, steroids, flavonoids, and essential oils \unskip~\cite{1021853:21553584}. The plant contains various medicinal properties and used in the treatment of skin diseases such as dermatitis, itching, scabies, leprosy, and chickenpox. 
    
\section{MATERIALS }
Fresh plant of \textit{Lantana camara} was collected from S.V. University, Thirupathi. Silver Nitrate purchased from Wise scientific, India. Gentamicin purchased from Aristo Pharmaceuticals, Mumbai, India. Cepftrione purchased from Ranbaxy Pharmaceuticals, Mohali, India. Diclofenac purchased from Cipla pharmaceuticals. India. All analytical grade and purchased from Hi Media (Mumbai). 

\textbf{Methods}

\textbf{Cold Extraction}

\textit{Lantana Camara} was separately shade dried, crushed to crude powder was macerated with ethanol and aqueous based on their polarity following cold extraction method, and filtrates were concentrated by rotary evaporator to yield was about 4\% dense residues \unskip~\cite{1021853:21553579}. Based on the phytochemical analysis and 99\% ethanol and aqueous extract were selected for further experiments. Ethanol and aqueous extract of \textit{Lantana camara} were prepared their anti microbial activity was done using microbial maximum inhibitory concentration.\ensuremath{^{}}

\textbf{Phytochemical Screening }

Test for flavonoids: Alkaline reagent\ensuremath{^{'}}s, Shinoda test

Test for alkaloids: Dragendroffs test, Mayers test, Wagner's test.

Test for carbohydrates: Benedict's test, Molischs test, Fehling's test

Test for proteins: Biuret test, Xanthoproteic test, Millon's test

Test for glycosides: Liebermann's test, Salkowskis test, Killer Killian test

Test for phenols: Ferric chloride test, Test for tannins, Gelatin solution

Test for free amino acids: Ninhydrin test, Test for quinines, HCL test; Test for the steroids: Liebermann Burchard test; Test for saponins: Foam test, Froth test \unskip~\cite{1021853:21553583}.

\textit{\textbf{In-vitro}} \textbf{ antibacterial activity}

\textbf{Disc diffusion method}

\textbf{Prepare the inoculums}

10\ensuremath{^{8 }} colonies forming unit per ml bacterial inoculums in a nutrient broth which is prepared by picking 3-5 isolated colonies from the plate \unskip~\cite{1021853:21553588}. The winning streak the swab on the surface going from nutrient agar go away for 5-10 minutes to dry the surface going from agar that permitting the bacteria to establish for themselves and on the media. Invert the plate and incubate them at 37\ensuremath{^{o}}C for 18-24 hours (Table~\ref{tw-ce0bd54eecac}).


\begin{table*}[!htbp]
\caption{\boldmath {Treatment schedule for \textit{In-vitro }antibacterial activity} }
\label{tw-ce0bd54eecac}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1615\linewidth-2\tabcolsep}p{\dimexpr.1585\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}p{\dimexpr.1525\linewidth-2\tabcolsep}p{\dimexpr.2075\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Groups & \multicolumn{4}{p{\dimexpr(.631\linewidth-2\tabcolsep)}}{\cAlignHack Treatment } & \\
\rowcolor{kwdboxcolor} & Media  & Inoculums & Antibiotic & Standard Dose  & Aqueous Extract of \textit{Lantana camara} Leaves AgNPs\\
\tblmidrule 
Normal  &
  Nutrient agar media &
  \textit{Staphylococcus aureus},  \mbox{}\protect\newline \textit{Escherichia coli.} &
  - &
  - &
  -\\
Standard  &
  Nutrient agar media &
  \textit{Staphylococcus aureus, } \mbox{}\protect\newline \textit{Escherichia coli.} &
  Gentamicin  &
  10mg/kg &
  -\\
Control  &
  Nutrient agar media &
  \textit{Staphylococcus aureus,} \mbox{}\protect\newline \textit{Escherichia coli} &
  - &
  - &
  -\\
Extract &
  Nutrient agar media &
  \textit{Staphylococcus aureus, } \mbox{}\protect\newline \textit{Escherichia coli} &
  - &
  - &
  10 mg/kg \\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Treatment schedule for \textit{In-vivo} studies of anti-bacterial activity} }
\label{tw-300d1260879c}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.21680000000000003\linewidth-2\tabcolsep}p{\dimexpr.7832000000000001\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Groups & \cAlignHack Treatment\\
\tblmidrule 
Group 1 &
  Animal served as the positive control (normal animals) were administered 0.9\% normal saline solution \\
Group 2 &
  Animal served as the negative control (disease animal) were administered 0.9\% normal saline solution\\
Group 3 &
  Animal treated with standard drug Gentamicin \\
Group 4 &
  Animals were administered graded doses of 200mg/kg bodyweight of the ethanol extract of \textit{Lantana camara} leaves respectively twice daily.\\
Group 5 &
  Animals were administered graded doses of 200mg/kg bodyweight of the Aqueous Extract of \textit{Lantana camara} leaves respectively twice daily.\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Treatment schedule for \textit{In-vivo} studies of anti-inflammatory activity} }
\label{tw-b2baadc7578b}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.17420000000000005\linewidth-2\tabcolsep}p{\dimexpr.8258000000000001\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Groups & \cAlignHack Treatment\\
\tblmidrule 
Group 1 &
  Animal served as the positive control (normal animals) were administered 0.9\% normal saline solution \\
Group 2 &
  Animal served as the negative control (disease animal) were administered 0.9\% normal saline solution\\
Group 3 &
  Animal treated with standard drugs Diclofenac.\\
Group 4 &
  Animals were administered graded doses of 200mg/kg bodyweight of the ethanol extract of \textit{Lantana camara} leaves respectively twice daily.\\
Group 5 &
  Animals were administered graded doses of 200mg/kg bodyweight of the Aqueous Extract of \textit{Lantana camara} leaves respectively twice daily. \\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Treatment schedule for \textit{In-vivo} studies of Wound-healing activity} }
\label{tw-431ce3b2d46f}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.17409999999999997\linewidth-2\tabcolsep}p{\dimexpr.8258999999999999\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Groups & \cAlignHack Treatment\\
\tblmidrule 
Group 1 &
  Animal served as the positive control (normal animals) were administered 0.9\% normal saline solution \\
Group 2 &
  Animal served as the negative control (disease animal) were administered 0.9\% normal saline solution\\
Group 3 &
  Animal treated with standard drugs Neosporin.\\
Group 4 &
  Animals were administered graded doses of 200mg/kg body weight of the ethanol extract of \textit{Lantana camara} leaves respectively twice daily.\\
Group 5 &
  Animals were administered graded doses of 200mg/kg body weight of the Aqueous extract of \textit{Lantana camara} leaves respectively twice daily.\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Phytochemical test of \textit{Lantana camara }Leaves} }
\label{tw-7149ce5e46a4}
\centering 
\begin{threeparttable}

\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.56370000000000005\linewidth-2\tabcolsep}p{\dimexpr.2244\linewidth-2\tabcolsep}p{\dimexpr.2119\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Phytochemical Constituents & Ethanol  & Aqueous\\
\tblmidrule 
\textbf{Flavonoids} &
   &
  \\
Alkaline reagent \mbox{}\protect\newline Shinoda test &
  + \mbox{}\protect\newline + &
  + \mbox{}\protect\newline +\\
\textbf{Alkaloids} &
   &
  \\
Wagner's test \mbox{}\protect\newline Mayers test &
  + \mbox{}\protect\newline + &
  - \mbox{}\protect\newline -\\
\textbf{ Carbohydrates} &
   &
  \\
Benedicts test \mbox{}\protect\newline Molischs test \mbox{}\protect\newline Fehling's test &
  + \mbox{}\protect\newline + \mbox{}\protect\newline + &
  + \mbox{}\protect\newline + \mbox{}\protect\newline +\\
\textbf{Proteins} &
   &
  \\
Biuret test \mbox{}\protect\newline Xanthoproteic test \mbox{}\protect\newline Millon's test &
  + \mbox{}\protect\newline + \mbox{}\protect\newline + &
  + \mbox{}\protect\newline + \mbox{}\protect\newline +\\
\textbf{Glycosides} &
   &
  \\
Liebermann's test \mbox{}\protect\newline Salkowskis test \mbox{}\protect\newline Keller Killian's test &
  + \mbox{}\protect\newline - \mbox{}\protect\newline + &
  - \mbox{}\protect\newline + \mbox{}\protect\newline -\\
\textbf{Phenols} &
   &
  \\
Ferric chloride test &
  + &
  +\\
\textbf{Tannins} &
   &
  \\
Gelatin test \mbox{}\protect\newline Braymer's test &
  - \mbox{}\protect\newline - &
  + \mbox{}\protect\newline +\\
\textbf{Amino acids } &
   &
  \\
Ninhydrin test &
  + &
  +\\
\textbf{Steroids} &
   &
  \\
Liebermann Burchard test  &
  + &
  -\\
\textbf{Quinines} &
   &
  \\
HCL test &
  - &
  -\\
\textbf{Saponins} &
   &
  \\
Foam test \mbox{}\protect\newline Froth test &
  - \mbox{}\protect\newline - &
  + \mbox{}\protect\newline +\\
\tblbottomrule 
\end{tabulary}\par 
\begin{tablenotes}\footnotesize 
    
\item{Note: (+) Present, (-) Absent}
\end{tablenotes}
\end{threeparttable}

\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Zone of inhibition of Ethanol Squeeze of \textit{Lantana camara} leaf } }
\label{tw-8a787fd73836}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.348\linewidth-2\tabcolsep}p{\dimexpr.312\linewidth-2\tabcolsep}p{\dimexpr.34\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Organisms & Gentamicin (2mg) & Ethanol Squeeze of leaf (2mg)\\
\tblmidrule 
\textit{Escherichia coli} &
  5.5 &
  3.5\\
\textit{Staphylococcus aureus} &
  6 &
  4.5\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {FTIR on \textit{Lantana camara} leaf extract of silver nanoparticles} }
\label{tw-aac713b038b9}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.0939\linewidth-2\tabcolsep}p{\dimexpr.29349999999999994\linewidth-2\tabcolsep}p{\dimexpr.2018\linewidth-2\tabcolsep}p{\dimexpr.182\linewidth-2\tabcolsep}p{\dimexpr.2288\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}S.No & Type of Vibration & \multicolumn{2}{p{\dimexpr(.38379999999999995\linewidth-2\tabcolsep)}}{\cAlignHack  Wave Number} & Possible Serious For Shift Alteration\\
\rowcolor{kwdboxcolor} &  & Plant Extract  & Nano Silver & \\
\tblmidrule 
1 &
  N-H stretching vibration of secondary amines &
  3334 &
  3329 &
  Chelation of N-H groups with silver\\
2 &
  O-H stretching vibration of carboxylic acids &
  2117 &
  2131 &
  Interaction of silver with OH group of flavonoids\\
3 &
  Stracting vibration of C=O aldehydes &
  1637 &
  1637 &
  Chelation of C=O groups with silver\\
4 &
  C-H grouping of phenyl ring substitution  &
  621 &
  581 &
  Interaction with phenyl ring or alkynes with silver\\
5 &
  C-H group of phenyl ring substitution or alkynes &
  524 &
  525 &
  C-H group of phenyl ring\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\bgroup
\fixFloatSize{images/2cc7f70d-dc32-4b84-9c32-0fb4464cc5ce-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2cc7f70d-dc32-4b84-9c32-0fb4464cc5ce-upicture1.png}{\includegraphics{images/2cc7f70d-dc32-4b84-9c32-0fb4464cc5ce-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Zone of inhibition of gentamicin, ethanol extract of \textit{Lantana camara}, gentamicin with LELC in \textit{E.coli }culture plate}}
\label{f-b4ce0975db65}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/e0569e44-745d-4f2b-b338-8cd529419056-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/e0569e44-745d-4f2b-b338-8cd529419056-upicture2.png}{\includegraphics{images/e0569e44-745d-4f2b-b338-8cd529419056-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Zone of inhibition of gentamicin, ethanol extract of \textit{Lantana camara}, gentamicin with ethanol extract of \textit{Lantana camara} in \textit{staphylococcus aureus} culture plate}}
\label{f-94de9884639b}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/1e00ca63-332c-406e-b89c-8ce7eed78485-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/1e00ca63-332c-406e-b89c-8ce7eed78485-upicture3.png}{\includegraphics{images/1e00ca63-332c-406e-b89c-8ce7eed78485-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Synthesized AgNPs solution several response time (A) 20 min (B) 70 min}}
\label{f-1f9c7bf1191c}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/6edee774-cdef-4921-945e-9bbae736deb2-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/6edee774-cdef-4921-945e-9bbae736deb2-upicture4.png}{\includegraphics{images/6edee774-cdef-4921-945e-9bbae736deb2-upicture4.png}}{}
\makeatother 
\caption{\boldmath {UV-Visible spectroscopy of \textit{Lantana camara} leaf extract of silver }}
\label{f-305878c6ac00}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d3113d0b-ca10-496d-95dd-f5dfae4a8b71-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d3113d0b-ca10-496d-95dd-f5dfae4a8b71-upicture5.png}{\includegraphics{images/d3113d0b-ca10-496d-95dd-f5dfae4a8b71-upicture5.png}}{}
\makeatother 
\caption{\boldmath {FTIR of \textit{Lantana camara} leaf extract}}
\label{f-1831489253f6}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/c550f6a2-8e1e-486f-b236-b661836abcc4-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c550f6a2-8e1e-486f-b236-b661836abcc4-upicture6.png}{\includegraphics{images/c550f6a2-8e1e-486f-b236-b661836abcc4-upicture6.png}}{}
\makeatother 
\caption{\boldmath {FTIR of \textit{Lantana camara} Silver Nanoparticles}}
\label{f-b54e595cddba}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/cfbc4a37-4749-4533-b0fa-180a797f5d0b-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/cfbc4a37-4749-4533-b0fa-180a797f5d0b-upicture7.png}{\includegraphics{images/cfbc4a37-4749-4533-b0fa-180a797f5d0b-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Particle size of \textit{Lantana camara} leaf extract of AgNPs}}
\label{f-592a59dcf324}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/713de47c-94f1-43c8-a8a1-d136f8bd5549-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/713de47c-94f1-43c8-a8a1-d136f8bd5549-upicture8.png}{\includegraphics{images/713de47c-94f1-43c8-a8a1-d136f8bd5549-upicture8.png}}{}
\makeatother 
\caption{\boldmath {Zeta potential of \textit{Lantana camara} leaf extract of silver Nanoparticles}}
\label{f-8df9f1781406}
\end{figure}
\egroup
\textit{\textbf{In-vivo}} \textbf{ Anti bacterial activity}

The mice were randomly divided into five groups (n=2), for the preparation of inoculating the bacteria on to nutrient agar and incubated at 37\ensuremath{^{o}}C overnight. Before inoculation, the mouse models of bacterial skin infection were sedated with either \unskip~\cite{1021853:21553587}. The flanks of the sedated mice were shaved with clippers when necessary and cleansed with an ethanol solution and then apply the bacterial solution on the skin. The treatment with antimicrobial used in this study begins after 4 hours of bacterial inoculation and continued at the regimens of 7 days (Table~\ref{tw-300d1260879c}).

\textit{\textbf{In vivo}} \textbf{ Method {\textendash} Anti inflammatory activity}

\textbf{Carrageen and induced paw edema in rats}

Ten healthy, domestic male albino rats, weighing 230-250 gm were used in this study. These mice were kept in separate cages the room temperature was maintained at 20-25\ensuremath{^{o}}C. Carrageenan-induced paw edema model was used for ethanol and aqueous extract of the \textit{Lantana camara} leaves on inflammation as shown in Table~\ref{tw-b2baadc7578b}. Acute inflammation was produced by the administration of 0.9\% mL of carrageenan in the sub plantar region of the left hind paw of a rat. The standard drug (diclofenac) and extract (50mg) were administered 30 min before the carrageenan injection. Inject to the control group and was treated with vehicle alone in the sub plantar region of the left hind paw of rats \unskip~\cite{1021853:21553578}. The standard drug was treated with diclofenac (50mg) used for the evaluation of the anti-inflammatory activity. The volume of the paw was measured immediately and also at the end of 2 hours and 4 hours after the administration of carrageenan using plethysmometer. The percentage inhibition of inflammation was calculated by the following formula:

$\style{font-family:'Times New Roman'}{\style{font-size:8px}{\begin{array}{l}\%\;Inhibition=\\\frac{Increase\;in\;paw\;volume\{(control)-(test)\}}{Increase\;in\;paw\;volume\;(Control)}\times10\end{array}}} $

\textit{\textbf{In-vivo}} \textbf{ Method (Wound-healing activity) }

The animals were randomly divided into two groups with a sample size of 5 rats/ group. Animals were ketamine (80mg) anesthetized before and during the creation of the wounds \unskip~\cite{1021853:21553585}. A full-thickness excision wound with a circular area of 200 mm and 2 mm depth was created along the markings using toothed forceps, a surgical blade, and pointed scissors. Animals were topical with the simple ointment (Neosporin) base as a placebo control Animals were treated topically with the 10\% ointment of the ethanol (200mg/kg daily) and aqueous(200mg/kg daily) extract of \textit{Lantana Camara} leaves till complete epithelisation as shown in Table~\ref{tw-431ce3b2d46f}. 

\textbf{Synthesis Consisting of Silver Nanoparticles on \textit{Lantana camara} Leaves Squeeze}

Ethanolic extract of \textit{Lantana camara} leaf 10 ml was added to the 90 ml of 1Mm AgNO\ensuremath{_{3}} a ratio of 9:1. This solution was kept at ambient room temperature and stirred continuously for 10 min using a magnetic stirrer \unskip~\cite{1021853:21553590}. After 24 hours changed green colour to dark brown colour which signifies the production of silver nanoparticles.

\textbf{Characterization of Silver nanoparticles}

\textbf{FT-IR analysis}

It turned into performed to spot the functional groups in \textit{Lantana Camara} that have been liable for the reduction of the silver nitrate and spectracular stabilization of silver nanoparticles \unskip~\cite{1021853:21553580}. \ensuremath{^{}}

\textbf{Particle size Analyzer}\ensuremath{^{}}

The basic study that offers information about the share out of the numerous smallest sizes of the particles spread in the stratified sample \unskip~\cite{1021853:21553582}.

\textbf{Zeta potential}

The surface attainable of silver nanoparticles this is fact the depiction of the stability of nanoparticles \unskip~\cite{1021853:21553586}.
    
\section{RESULTS AND DISCUSSION}
\textbf{Phytochemical Screening}

Ethanolic extract and aqueous extract of \textit{Lantana camara} Leaf extract were screened and the data was given in Table~\ref{tw-7149ce5e46a4}.

\textit{\textbf{In-vitro}} \textbf{ anti-Bacterial Activity}

\textbf{Disc diffusion method}

Ethanol extract of \textit{Lantana camara}, when given for \textit{E.coli} bacteria, shows a zone of inhibition of 4.5mm, 3.5mm for gentamicin (2mg), ethanol extract of \textit{Lantana Camara} (2mg) respectively (Figure~\ref{f-b4ce0975db65}).

Ethanol extract of \textit{Lantana camara}, when given for \textit{Staphylococcus aureus} bacteria, shows a zone of inhibition of 6mm, 4.5mm for gentamicin (2mg), ethanol extract of \textit{Lantana camara} (2mg) respectively (Figure~\ref{f-94de9884639b})

Zone of inhibition for gentamicin, was slightly more compound to ethanol extract of \textit{Lantana camara} leaf which is given in Table~\ref{tw-8a787fd73836}.

From the data of phytochemical screening and \textit{in-vitro} antibacterial utilisation, it is revealed that the liquid extract of \textit{Lantana camara} leaves shows more constituents and good anti-bacterial activity compared to ethanol extract.

So it turned into considered so the aqueous extract of leaves as promising extract which was formulated as silver nanoparticles to enhance elegance, acceptability and bio-availability. 

\textbf{Characterization of Silver nanoparticles}

\textbf{UV-Vis spectroscopy}

A visible colour change delight in see through to brownness inside of 20 minutes signifies the more formation of AgNPs, which was confirmed by UV-Visible spectrometer (Figure~\ref{f-305878c6ac00}). After 70 minutes there has been a vital colour to dark brown due to increased reaction time which reinforces the growth of silver nanoparticles (Figure~\ref{f-1f9c7bf1191c}). 

The intensity of the absorption peak at 418 nm increased with an increasing period of the aqueous component. \textbf{\space }

\textbf{Nanoparticles}

\textbf{FT-IR analysis}

The infrared Spectrum used to be analyzing the functional grouping present in \textit{Lantana Camara} leaf extract (Figures~\ref{f-1831489253f6} and~\ref{f-b54e595cddba}). The FTIR spectrums of \textit{Lantana camara} leaf extract and silver nanoparticles are represented (Table~\ref{tw-aac713b038b9}). 

\textbf{Particle size Analyzer}

Partial size determination of \textit{Lantana camara} leaf the synthesized AgNPs were shown by the intensity and laser diffraction revealed that particles obtained are a polydisperse mixture with the size ranging from 10 to 30nm. The average diameter of the particles was found to be leaf 2657.2nm (Figure~\ref{f-592a59dcf324}).

\textbf{Zeta potential}

The \textit{Lantana camara} leaf sequenced silver nanoparticles are decided in water as a dispersant. it encounter to leaf 0.3mV (Figure~\ref{f-8df9f1781406}).

Plants contain many bioactive chemical constituents with various pharmacological activities. Many potent and effective medicinal properties used for treating fatal diseases have been isolated from various medicinal plants. Hence the demand for medicinal herbal drugs is increasing day by day. The study of the medicinal plants makes miracles in the medical field. \textit{Lantana camara} was selected for the present study.

The anti-microbial properties of the plant have been examined by numerous researchers worldwide because of the therapeutic potential accredited to \textit{Lantana Camara}. Collection of \textit{Lantana camara} leaves and preparation of aqueous and ethanol on the cold extraction method. Further Preliminary phytochemical screening of \textit{Lantana camara} extracts to identify phytoconstituents. This study can be used in the traditional medicinal system to cure diseases and therapeutic use. The interaction going from aqueous slot with the silver nitrate solution and showed modernized deepening of color into change yellow color to dark brown color.
    
\section{CONCLUSION}
The present research demonstrates that phytochemical constituents of ethanol and aqueous extract of \textit{Lantana camara} leaves and phytostabilized of AgNPs can act as a unique of antibacterial activity against of organisms like \textit{staphylococcus aureus} and \textit{Escherichia coli } responsible for infections in burns and \textit{In-vivo } methods of \textit{Lantana camara} leaves extract of antibacterial activity, antiinflammatory activity and wound healing activity. The personalities of water-soluble flavonoids in plant organ extract were responsible for spectracular reduction process of silver nanoparticles. Anti-bacterial activities of ethanol extract of \textit{Lantana camara} were carried out and it shows that the antibacterial activity of ethanol extract of \textit{Lantana Camara} is slightly less compared to gentamicin and ceftriaxone. Zone of inhibition for gentamicin, ceftriaxone was slightly more compound to ethanol extract of \textit{Lantana camara} leaf.
    
\section{Acknowledgement}
The authors are heartily thankful to Dr. C. Appa Rao M. Pharm, Ph. D., Principal, Department of Bio-chemistry, Sri Venkateswara University of pharmaceutical sciences, Tirupati-517501, Andhra Pradesh, India for permitting to do the work and providing all the necessary facilities. 

\textbf{Conflict of interest}

The authors attest that they have no conflict of interest in this study.

\textbf{Funding support}

No financial support for the current study.
    

\bibliographystyle{unsrt}

\bibliography{\jobname}\vspace*{20pt} 
\begin{mdframed}[backgroundcolor=cpboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=1pt,rightmargin=1pt,innerleftmargin=1pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Copyright:}~\copyrightstmt}	
\end{mdframed}
        
\begin{mdframed}[backgroundcolor=citeboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=4pt,rightmargin=4pt,innerleftmargin=2pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=4pt,skipbelow=2pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Cite~this~article:}~\citeauthor.~\citetitle.\space\makeatletter\abbrvJournalTitle\ \@copyrightYear;\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi\ifx\@journalIssue\@empty\else(\@journalIssue)\fi:\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}.\makeatother}	
\end{mdframed}
\vskip-1.1pc

\vskip.5pc
{\includegraphics[width=5cm]{pharmasprings_logo.png}}
~\\
{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \makeatletter\@copyrightYear\makeatother\ Pharma Springs Publication.}

\end{document}
